Author’s response to reviews

Title: Measuring dementia carers’ unmet need for services - an exploratory mixed method study.

Authors:

Christine M Stirling (Christine.Stirling@utas.edu.au)
Sharon Andrews (Sharon.Andrews@utas.edu.au)
Toby Croft (toby.croft@dhhs.tas.gov.au)
James Vickers (James.Vickers@utas.edu.au)
Paul Turner (paul.turner@utas.edu.au)
Andrew Robinson (Andrew.Robinson@utas.edu.au)

Version: 3 Date: 20 April 2010

Author’s response to reviews:

To: Editor BMC Health Services Research
6/24/2009

Please accept this revised paper titled “Measuring Dementia Carers’ Unmet Need for Services – an exploratory mixed method study” for publication with your journal BMC Health Services Research. The paper has been checked to ensure it matches the style requirements of your journal and suggested changes have been made, as below.

Response to Reviewers:
The term ‘dementia sufferers’ has been deleted and replaced with the term people with dementia.

There were no other suggested changes.

Yours Sincerely
Corresponding Author

Dr Christine Stirling
Senior Research Fellow
Wicking Dementia Research and Education Centre